Author’s response to reviews

Title: Geographic surveillance of community associated MRSA infections in children using electronic health record data

Authors:

Lilly Immergluck (limmergluck@msm.edu)

Traci Leong (tleong@emory.edu)

Kevin Matthews (yrp4@cdc.gov)

Khusdeep Malhotra (tuf81845@temple.edu)

Trisha Parker (trisha.parker@choa.org)

Fatima Ali (f.imanali@gmail.com)

Robert Jerris (robert.jerris@choa.org)

George Rust (george.rust@med.fsu.edu)

Version: 4 Date: 28 Dec 2018

Author’s response to reviews:

Technical Comments:

1. We note that two authors have the same initials (KM). To distinguish between them in the Authors’ Contributions section please designate them as KM1 and KM2, with KM1 corresponding to Kevin Matthews, chosen because this name is furthest up the author list.

Authors’ response: Thank you for this suggestion and we have changed the initials in the ‘Authors’ Contributions’ section.

KM1 is Khusdeep Malhotra

KM2 is Kevin Matthews

2. In your Competing Interests section, please state whether or not the authors have any competing interests.

Authors’ response: The authors have no competing or conflict of interests and each have sent me an email to state this.
3. Please remove the Conflict of Interest section from the Declarations.

Authors’ response: Thank you. We have removed this section from the ‘Declarations’ section.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors’ response: Thank you. We have uploaded a final clean version that does not contain any tracked changes or comments.